Discussion Paper

Research Department, Central Bureau of Statistics, Norway

No. 96

July 1993

An Empirical Life Cycle Model of Savings, Labour Supply and Consumption without Intertemporal Separability

by

Tom Kornstad

An Empirical Life Cycle Model of Savings, Labour Supply and Consumption without Intertemporal Separability

by

Tom Kornstad

Abstract

This paper formulates and estimates a structural life cycle model of married couples' labour supply and consumption of durables and non-durables. The purpose of this work has been to find a specification of this class of life cycle models that can be estimated in the absence of observations of consumption of non-durables, and the price and the physical stock of durables. We allow for a particular kind of non-separability in the demand for durables, and treat durables and non-durables as a (single) Hicks composite good.

Keywords: Empirical, life cycle model, labour supply, consumption, intertemporal non-separability

Acknowledgement: I wish to thank my supervisors Steinar Strøm, and in particular John K. Dagsvik, for their advice all through my dr. polit. dissertation research. I am also grateful to Margaret Simpson for her detailed comments on this paper. The research has been financially supported by the Economic Research Programme on Taxation from The Norwegian Research Council for Applied Social Science.

1 Introduction

This paper formulates and estimates an empirical life cycle model for two-person households that choose labour supply and consumption including consumption of durables in an environment of uncertainty about future prices. Ideally, estimation of this model requires complete life cycle data for all these goods and households' expectations about the corresponding prices, including the interest and the income tax rates. Since there exist no such data sets, and economic theory is rather weak on how to measure the demand for durables, empirical work treats at least some of the goods in a rather summary way. We will also argue that measurement error in the observations of the consumption of non-durables may be important. The point of departure of this work has then been to modify the theory so that a complete life cycle model of labour supply and consumption of durables and non-durables can be estimated in the absence of reliable data for consumption of non-durables, and the purchase price and the physical stock of durables. In contrast to the standard, and presumably unrealistic, assumption of intertemporal separability, we allow for a particular kind of non-separability in the demand for durables.

The paper is organized as follows. Section 2 presents the standard life cycle model of consumption of durables and non-durables, and discusses some of the problems related to estimation of preferences over these goods. We discuss the serious measurement problems, and raise the important point that the first order condition for durables does not reflect the existence of fixed costs in the demand for durables. The standard assumption of intertemporal separability is also discussed.

In Section 3 we present the general framework of the complete model to be estimated. We introduce a particular form of habit persistence in the demand for durables. The specification is analogous to the one that is found in Bover [5], Hotz, Kydland and Sedlacek [14], Johnson and Pencavel [16] and Muellbauer [24]. However, in contrast to these works, which analyse labour supply and consumption of non-durables, we relate this specification to the stock of durables. We argue that

our specification is consistent with an interpretation that says that the household gains utility from an increase in the stock of durables relative to what "remains" from the last period. In the determination of what "remains" from the last period, we allow for physical as well as psychological depreciation. Psychological depreciation is introduced to take into account that after some time the household may tire of the present stock of durables. This means the physically existing capital stock does not yield as much utility as before, but since it can be resold in the market, it must be treated differently from the stock that is physically depreciated.

If the stock of durables is totally depreciated (in the utility context), our specification of the utility of durables implies that utility depends on the stock of durables. Hence, the standard specification of the utility of durables is a special case of our specification.

It appears that an advantage of our interpretation is that it leads us towards a specification that can be estimated in the absence of reliable observations of the consumption of non-durables, and the purchase price and the physical stock of durables. This advantage is demonstrated, before we end section 3 by commenting on the estimation strategy. Section 4 presents a particular empirical specification which is estimated, and in Section 5 we look at the details of the estimation procedure. Section 6 gives an account of the data, and Section 7 presents the estimation results. Finally, some concluding remarks are made in Section 8.

2 Confronting the Standard Life Cycle Theory with Data

2.1 The standard framework

We now look at some problems related to the estimation of preferences over durables and non-durables. In order to focus on the main problems, assume that lifetime preferences are an inter- and intratemporal separable function of durables and ordinary consumer goods only. Thus

$$U_0 = \sum_{t=0}^{T} (1+\rho)^{-t} \left[U_{ct}(C_t) + U_{kt}(K_t) \right], \tag{1}$$

where ρ is the time preference rate, K denotes stock of durables, C is consumption of non-durables, and subscript 0 and T denote the beginning and the end of the planning horizon. We also assume perfect certainty about future prices and preferences, and that labour supply, H, is exogenous.

In the absence of personal income taxation, the within period wealth constraints are

$$w_t H_t + (1 + r_t) F_{t-1} = p_t C_t + q_t [K_t - (1 - \delta_f) K_{t-1}] + F_t, \quad t = 0, 1, \dots, T, \quad (2)$$

where F is assets measured in nominal terms, r is the nominal interest rate, w, p and q are the nominal prices of leisure, non-durables and durables, and δ_f is the physical depreciation rate for durables. If $\delta_f < 1$ (and ≥ 0), the specification of the wealth constraint implies that durables can be sold in a second hand market.

Despite the fact that households may face different¹ prices, we assume that all (pre-tax) prices are equal for all households. Except for the fact that bequests at the beginning of the planning period can be exogenously included into the value of F_0 , we do not take explicit account of the possibility of bequests. Neither do we consider that families can place their wealth into other kinds of assets, such as bonds, shares, pension funds, arts or antiques.

We assume that durables are bought at the end of the period. Since this stock is continuously being depreciated during the subsequent period, $q_t(1-\delta_f)K_{t-1}$ is the market value of the stock of durables that was demanded in the last period and that is left at the end of period t.

In what follows we assume that there are no binding credit market restrictions. Maximization of the preference function (1) subject to the wealth constraint (2),

¹The households may face different prices, for example, because they live in different regions of the country or because banks offer different terms to different customers.

given values of initial stocks of assets and durables, and the terminal stock of assets, yields the following first-order conditions,

$$\frac{\partial U_{ct}}{\partial C_{\bullet}} = \lambda_t p_t, \quad t = 0, 1, \dots, T, \tag{3}$$

$$\frac{\partial U_{kt}}{\partial K_t} = \lambda_t q_t - \frac{1}{1+\rho} \lambda_{t+1} (1-\delta_f) q_{t+1}, \quad t = 0, 1, \dots, T, \tag{4}$$

and

$$\lambda_t = \frac{1}{1+\rho} (1+r_{t+1}) \lambda_{t+1}, \quad t = 0, 1, \dots, T-1, \tag{5}$$

where λ_t is the Lagrange multiplier, i.e. the marginal utility of wealth.

2.2 Some data - and measurement problems

Equations (3) to (5) and the wealth constraints (2) constitute a simultaneous equation system that determines C_t , K_t , F_t and λ_t , for $t=0,1,\cdots,T$, as functions of all the exogenous variables of the model, including the parameters of the lifetime utility index. In most cases it is, however, impossible to find these reduced form equations. If λ can be observed, however, it appears that the relevant demand functions in the life cycle context are the Frisch demands, cf. MaCurdy [20].

Concerning the estimation of these functions, it is a problem that λ_t is latent and depends on all the exogenous variables of model. Kornstad [18] discusses various solutions to this problem. These solutions typically involve a two stage estimation procedure, where the first stage involves estimation of the parameters related to intraperiod allocations, while the second stage involves estimation of the parameters related to interperiod allocations. However, even if one estimates only within period preferences (stage one), which typically requires less data, it appears that the data requirement is quite large. For instance, in the estimation of the differenced marginal utility of wealth constant demand functions, MaCurdy's fixed effect approach, cf. [20, 22], demands panel data for consumption and (after-tax) prices for all goods (at least one), for at least two periods. Similarly, estimation of the y-conditional demands in the two stage budgeting theory discussed in Blundell [4]

and Blundell and Walker [3], demands cross section data for the consumption of all goods, for a variable labelled full income, and for the simultaneous distribution of all current (after-tax) prices. Altonji [2] and MaCurdy [21] propose using the marginal rate of substitution functions, and this approach requires cross section data for all goods (at least two) and their (after-tax) prices. In addition, these approaches may need instrument and taste-modifier variables, and in order to account for eventual cohort effects, estimation of the instrument equations may require panel data. Since the decisions with respect to labour supply and consumption of non-durables and durables are taken simultaneously, ideally, our data should include observations for all these goods. Taking also into consideration the fact that estimation of the parameters determining interperiod allocations typically requires panel data, estimation of complete life cycle models of labour supply and consumption is very data demanding. This extensive data requirement is important to have in mind as we look at some problems that are particular to the estimation of the preferences over durables and non-durables.

A particular problem in the estimation of equation (3), is the measurement of consumption of non-durables. Non-durables include a variety of goods, and the measurement of the consumption of all these goods requires detailed household accounts. To reduce costs, many goods are bought infrequently, but in large quantities each time. This fact, and the fact that the consumption of many goods is season specific, implies that consumption patterns should be observed over longer time periods. Hence, without having access to detailed household accounts for longer time periods, it is hard to obtain reliable observations of the consumption of non-durables.

However, if we observe the cash flow related to the purchases of durables, we may use the period specific wealth constraints and the consumer price index to calculate the consumption of non-durables. But since we, in most cases, do not observe the purchase of durables in addition to all the other variables we need, this method is rarely useful.

Concerning the demand for durables, notice that if there are no binding con-

straints in the credit market, the first order condition (4) can be written

$$\frac{\partial U_{ct}}{\partial K_t} = \lambda_t \left[q_t - (1 - \delta_f) \frac{q_{t+1}}{1 + r_{t+1}} \right], \tag{6}$$

where $q_t - [(1 - \delta_f)q_{t+1}/(1 + r_{t+1})]$ is the user price of durables (including capital gains).

Again, measurement problems are serious. According to economic theory, consumption should be measured in physical units or in real terms, and not as an expenditure evaluated in current prices. It is also evident that consumption is a stream which should be measured per time unit, but apart from that economic theory gives very few guidelines. Since durables yield a flow of services for many periods, it is difficult to quantify the consumption of durables for a particular year.

Assuming there is a fixed relationship between the consumption and the stock of durables, the stock can be used as an argument in the utility index, cf. equation (1). The measurement problems, however, are still serious. An important reason is that durables typically have many characteristics that influence utility. For example, the utility of housing depends on characteristics such as location, the number of different kinds of rooms and "quality". The multiplicity of characteristics mean that a division of housing into homogenous subgroups will inevitably result in a huge number of groups. Even if it was possible to define these groups, it is still a problem that no existing data set includes all the information required to determine the distribution of all housing into the various groups. This fact becomes even more accentuated when we consider that estimation typically requires observations of other variables as well.

According to equation (6), the user price of durables is the relevant price for the demand for durables. Another problem related to the estimation of preferences over durables then is how to measure the user price of durables. This price is only rarely realized in the market, which means that for practical estimation purposes it cannot be observed directly. An alternative is to calculate it indirectly using observations of the purchase prices, the (after-tax) interest rate and the physical depreciation rate.

Notice then that the practical problems related to the measurement of durables also make it problematic to quantify the purchase price of durables. Since the user price includes the purchase price from two different periods, and the depreciation rate is rarely known, this means that the indirect method can not be used either.

The difficulty of observing the purchase price of durables suggests an approximation in which we use the (real) market value of durables² as an argument of the utility index. A problem with this approach is that, since most second-hand durables are only rarely resold, it is difficult to obtain reliable predictions of the market values for all relevant goods. Although this problem may be serious, it is not obvious that it is more serious that the problems related to the measurement of expenditures on non-durables. It may be easier to ask for information about the most important durables than about a large number of non-durables.

The fact that many data sets do not include the market value of durables, in addition to all the other variables we need for estimation, is a serious problem. Since we typically do not observe $q_t(1-\delta_f)K_{t-1}$ and the other components of the wealth constraint, indirect observation through the wealth constraint is also difficult.

We therefore conclude that it is difficult to obtain reliable observations on the market value of durables. Due to missing theoretical foundations, it is even more difficult to determine and observe their consumption and prices. Reliable observations of the consumption of non-durables are also difficult to obtain, but this is due to practical problems and cost considerations, rather than theoretical problems.

In what follows we assume that the only consumption data we observe are $q_t K_{t-1}$ and the total cash flow related to the purchases of durables and non-durables. The cash flow may be calculated indirectly through the wealth constraints.

²In order to get a kind of a real value the market value can eventually be divided with the consumer price index.

2.3 Some additional arguments in favour of an alternative specification for durables

If δ_f equals one, either because there is no second-hand market or because physical depreciation is high, the user price coincides with the purchase price. Since durables and non-durables are treated in the same way in the utility index, this means that, apart from the fact that some durables can be resold, standard economic theory treats durables analogously to non-durables. As will now be argued, it is not obvious that this should be the case.

Standard life cycle theory assumes that consumption of durables can be treated as a continuous variable, and that no transactions costs are incurred in purchasing these goods. According to the first order condition (4), consumption should be adjusted until the marginal utility of durables equals its (net) marginal cost. This means that the demand for durables should continuously be adjusted in accordance with changes in the user price. In the case of housing, these adjustments involve important transactions costs of both the pecuniary and non-pecuniary kinds. If we divide housing into homogeneous subgroups with respect to location, number of rooms, and quality, we can also question whether it is reasonable to assume that it is possible to make marginal changes in its consumption. (This fact is also relevant for labour supply.) Both these facts raise the question of whether the marginal, and quite fine, comparisons that follow from the first order conditions, are good approximations to the comparisons households actually make. Maybe the household adjusts more roughly, in the sense that it can only choose between a limited number of baskets of goods. In the case of the demand for housing we can think of this basket as consisting of the purchase price of the housing, location, number of rooms, and a measure of quality. Households then choose the basket that maximizes utility.

Another argument in favour of a specification for durables other than the one presented in section 2.1 is the following. Economic theory allows $\partial(\partial U_t/\partial X_t)/\partial X_{t'}^*$ $\neq 0$, where X_t and $X_t^* = C_t, K_t$, for $t, t' = 0, 1, 2, \dots T$. That is, the marginal

utilities may depend on the consumption of all goods in all periods. Estimation of specifications that are based on the first order conditions then requires life cycle data for the consumption of all goods (in addition to the other variables that are needed).

Given this fact, most empirical analyses realize that some kind of separability assumption must be invoked. Most empirical analyses assume that preferences can be reasonably approximated by an intertemporal separable function, and it is also frequently assumed that even the within-period utilities are separable, cf. equation (1).

Intertemporal separability means that $\partial(\partial U_t/\partial X_t)/\partial X_{t'}^* = 0$, where X_t and $X_t^* = C_t$, K_t , $t \neq t'$ for $t, t' = 0, 1, 2, \dots T$. That is, the marginal utilities are functions of current consumption only, and it is possible to estimate within-period preferences from cross sectional data only, cf. Altonji [2], Blundell [4] and MaCurdy [21].

Intertemporal separability does, however, imply that habits play no role in the demand for the various goods. Duesenberry's relative income hypothesis (1949) focuses on habits in aggregate demand, and Muellbauer [24] claims that "Evidence from the estimation of complete systems of demand equations suggests that habits or persistence effects play an important role in consumer behaviour.". Bover [5] concludes that her estimation of a life cycle model "strongly support the importance of past hours in determining current hours decisions.". We therefore conclude that the assumption of intertemporal separability seems to be rather unrealistic.

3 A Model with Habits or Psychological Depreciation

This section considers modifying the life cycle model outlined in the previous section to account for habits in the demand for durables. The modification relies on a story in which habits inference the demand for durables, but not the demand for nondurables. An advantage of this modification is that it can be estimated for the case in which we observe neither the consumption of non-durables nor the price and the physical stock of durables.

The model is extended to include female and male labour supply, personal income taxation and the possibility of credit market constraints, and constitutes the general framework of the empirical analysis.

3.1 The model formulation

Assume lifetime preferences

$$U_{t} = \sum_{k=t}^{T} \frac{1}{(1+\rho)^{k-t}} G \left[U_{zk}(Z_{k}) + \sum_{j=f,m} U_{jk}(L_{jk}) \right], \tag{7}$$

where L_f and L_m are female and male leisure, and Z is a composite good defined as

$$Z_{k} \equiv C_{k} + \frac{q_{k}}{p_{k}} \left[K_{k} - \beta K_{k-1} \right], \tag{8}$$

where β is a constant. The good Z then includes both non-durables and durables. In the next section we present a theory that leads to this specification and the interpretation of β , but for the moment we merely postulate that Z is the relevant consumption good related to the consumption of durables and non-durables.

In order to account for personal income taxation, we introduce the tax function $I_k(w_{fk}H_{fk}, w_{mk}H_{mk}, r_kF_{k-1})$. Thus, income taxes are levied on wage and interest incomes, and we ignore wealth taxes. If F is negative, the household is in a net debt position, and the specification of the tax function allows deductions of interest expenses. The index k on the tax function indicates that the marginal tax rates vary across periods.

The time invariant monotonic transformation of within-period utility, G, implies that we allow for a particular kind of non-separability in the within-period utility of Z, L_f and L_m , cf. equation (13) and (14). It can be argued that this transformation does not reflect the special dependency between the household's utility of female and

male leisure. An alternative would then be to introduce a separate transformation of $[U_{fk}(L_{fk}) + U_{mk}(L_{mk})]$. Given the proposed estimation procedure, cf. equation (36) and (37), we cannot, however, use a similar Box-Cox transformation of $[U_{fk}(L_{fk}) + U_{mk}(L_{mk})]$ as we have used for $[U_{zk}(Z_k) + U_{fk}(L_{fk}) + U_{mk}(L_{mk})]$, cf. equation (31).

The period specific wealth constraints corresponding to the utility index (7) now become

$$\sum_{j=f,m} w_{jk} H_{jk} + r_k F_{k-1} = I_k \left(w_{fk} H_{fk}, w_{mk} H_{mk}, r_k F_{k-1} \right) + p_k Z_k - \delta_p q_k K_{k-1} + \Delta F_k, \quad k = t, \dots, T.$$
 (9)

Since purchase of durables are often financed by loan, the possibility of credit market constraints should be considered. We follow Mariger [23], and assume that the families can borrow only against mortgage in property. That is, net debt raised at the end of the period, $-F_k$, cannot exceed a fraction κ of the market value of the stock of durables at the end of that period, $q_k K_k$,

$$-F_{k} \leq \kappa q_{k} K_{k}, \quad k = t, t + 1, \cdots, T. \tag{10}$$

It is assumed that κ is invariant of both time and characteristics of the household.

As opposed to Mariger, we do not observe whether or not a household is constrained since we do not know κ and do not observe $q_k K_k$. (In the empirical analysis we assume κ is known and approximate $q_k K_k$ by $q_{k+1} K_k$.)

Despite the importance of labour market constraints we only account for the non-negativity constraint,

$$H_{jk} \ge 0, \quad j = f, m, \quad k = t, t + 1, \dots, T.$$
 (11)

Leisure and labour supply are also constrained through the time budget, that is

$$L_{jk} = \overline{L} - H_{jk}, \quad k = t, t + 1, \dots, T.$$
 (12)

We also allow for uncertainty about future prices, including the marginal tax and interest rates. The household maximizes the expected value of the time-preference-discounted sum of lifetime preferences with respect to $Z_{t'}$, $H_{ft'}$ and $H_{mt'}$, for t' =

 $t, t+1, \dots, T$, subject to the constraints stated above and given values of F_{t-1} , K_{t-1} and F_T .

Necessary conditions for an optimum are the satisfaction of all the constraints and

$$\tilde{G}' \frac{\partial \tilde{U}_{zt}}{\partial Z_t} = \lambda_t p_t, \tag{13}$$

$$\tilde{G}' \frac{\partial \tilde{U}_{jt}}{\partial L_{jt}} = \lambda_t \tilde{m}_{jt} + \alpha_{jt}, \quad j = f, m$$
 (14)

and

$$\lambda_{t} = \frac{1}{1+\rho} E_{t} \left[(1 + \tilde{R}_{t+1}) \lambda_{t+1} \right] + \psi_{t}, \tag{15}$$

where "tilde" denotes that the variable is evaluated at the optimum, G' is the partial derivative of G with respect to $[U_{zt}(Z_t) + \sum U_{jt}(L_{jt})]$, and

$$m_{jt} \equiv w_{jt} \left[1 - \frac{\partial I_t}{\partial (w_{jt} H_{jt})} \right] \tag{16}$$

and

$$R_{t+1} \equiv r_{t+1} \left[1 - \frac{\partial I_{t+1}}{\partial (r_{t+1} F_t)} \right] \tag{17}$$

are the after-tax marginal wage and interest rates. The Lagrange multipliers λ_t , ψ_t and α_{jt} , j = f, m, which are associated with the wealth constraint, the borrowing constraint and the labour supply constraints, are all household- or person-specific.

3.2 The specification of the composite good

To focus on the essential properties of the composite good, neglect the monotonic transformation function G of the preference function (7), and assume that the within-period preferences over consumption of non-durables and durables can be formulated as

$$V_{0} = V_{0}(K_{0}, K_{1}, \dots, K_{T}, C_{0}, C_{1}, \dots, C_{T}) = \sum_{t=0}^{T} \frac{1}{(1+\rho)^{t}} \left\{ U_{kt} \left[K_{t} - \beta K_{t-1} \right] + U_{ct}(C_{t}) \right\}.$$
(18)

Concerning our specification of the utility of durables, we notice that Bover [5], Hotz, Kydland and Sedlacek [14], Johnson and Pencavel [16] and Muellbauer [24] use a similar specification of the utility of leisure and non-durables.

Notice also that if $\beta = 0$,

$$V_0 = \sum_{t=0}^{T} \frac{1}{(1+\rho)^t} \left\{ U_{kt}(K_t) + U_{ct}(C_t) \right\}.$$
 (19)

That is, if $\beta = 0$, our treatment of durables coincides with the standard specification (1). In order to introduce more flexibility in the demand for durables, we assume that the preference structure underlying the demand behaviour can be defined with respect to a reference bundle, cf. Spinnewyn [27], Phlips [25, 26], Houthakker and Taylor [15], and Stadt, Kapteyn and Geer [28].

The choice of reference bundle varies across analyses. Houthakker and Taylor [15] assume that when households hold physical inventory of the actual good, the demand for the good depends on its stock. The reference bundle is now the stock of the good. For other goods such as tobacco, beer and spirits households do not hold inventories of any significance. Consumption of these stimulus are, however, habit-forming, and we can say metaphorically that households have built up a psychological stock of smoking and drinking habits, cf. Houthakker and Taylor. The reference bundle is now this psychological stock of habits, and according to Phlips [26], this "built-in" mechanism is typical for other kinds of non-durables as well.

While these two examples are related to habit formation, Alessie and Kapteyn [1] are concerned with the fact that household preferences are influenced by the behaviour of other groups of households. The reference bundle is then the consumption of these groups, but the reason for introducing the reference bundle is not habit formation. Phlips [26] classifies this type of taste changes as taste changes that result from better outside information due to external influences on the household. For instance, by talking with its neighbours, or reading advertising, the household may obtain new knowledge about goods that it has not yet purchased. Another example is that the consumption of a particular good may reveal that the good is harmful to

the health of the consumer.

The reference bundle can also represent minimal requirements of different goods from a physical point of view, cf. the linear expenditure system.

These examples indicate that the reference bundle may well change over time, and that it may depend on past experience. According to the habit-interpretation, we would expect that β (in equation (18)) satisfies the condition $0 < \beta \le 1$, and that β measures the extent to which habits influence current decisions³. Notice also that a more general means of accounting for habits, would be to introduce $\sum_{j=1}^{J} \beta_j K_{t-j}$, where we may reasonably assume $\beta_{t-m} > \beta_{t-n}$, for n > m.

Specification (18) implies that households have rational, in contrast to myopic, habits, cf. Muellbauer [24] and Spinnewyn [27]. This means that households recognise that current consumption decisions influence their future marginal rates of substitution. A disadvantage of assuming rational habits from an econometric point of view is that, if households maximize utility with respect to K, the two-stage budgeting property no longer holds since the intertemporal utility function is no longer separable. In what follows we will assume that the relevant good related to the consumption of durables can be measured by $(K_t - \beta K_{t-1})$, and in this case intertemporal utility is separable (in $(K_t - \beta K_{t-1})$).

In order to present an interpretation of β which is particularly relevant for durables, split β into two components according to

$$\beta \equiv 1 - \delta_f - \delta_p, \tag{20}$$

where we recall that δ_f is the parameter for physical depreciation, and where δ_p is the parameter we label as psychological depreciation.

It is assumed that habits are determined by the stock of durables that was demanded in the last period after deductions for physical as well as psychological

³Muellbauer [24], using quarterly US consumption data, claims that β can be negative if K_t is purchases of durable goods, but this argument does not fit our specification since K_t is stock of durables.

depreciation. Physical depreciation equals $\delta_f K_{t-1}$. The reason for introducing psychological depreciation is that, according to our theory, the household gets tired of the current stock of durables and wants a change. An example may be the household that moves into a new flat and after some time starts looking for a new flat. Another example may be the husband who, for some months, is very fascinated with his new car, but after some time loses interest. In a utility context this depreciation should also be deducted. But since the psychologically depreciated good $\delta_p K_{t-1}$ can be sold in the market, we must distinguish between physical and psychological depreciation. At the end of period t the utility relevant stock of durables is then $(1 - \delta_f - \delta_p) K_{t-1}$.

According to this story it seems reasonable to assume that $\delta_p \geq 0$. On the other hand, if durables are not worthless with respect to utility, but for some reason or another are of sentimental value, δ_p can well be negative. It is then reasonable to assume that the size of δ_p depends on the actual good, and that it may be household-and age-specific. Despite these facts, in the empirical analysis we approximate and assume that δ_p is constant, that is, independent of all these variables.

The upper limit of β implies that $\delta_f + \delta_p \geq 0$. If $\delta_f + \delta_p = 0$, equation (18) reduces to

$$V_0 = \sum_{t=0}^{T} \frac{1}{(1+\rho)^t} \left\{ U_{kt} \left[K_t - K_{t-1} \right] + U_{ct}(C_t) \right\}. \tag{21}$$

This means that for the special case that $\beta = 1$, equation (18) implies that the household is concerned about the increase in the stock of durables.

It remains to explain the relationship between equation (7) and (18) in their specification of preferences over non-durables and durables. By substituting $(K_k - \beta K_{k-1})$ into the wealth constraint (9), the wealth constraint can be written

$$\sum_{j=f,m} w_{jk} H_{jk} + r_k F_{k-1} - I_k \left(w_{fk} H_{fk}, w_{mk} H_{mk}, r_k F_{k-1} \right) = \tag{22}$$

$$p_k C_k + q_k [K_k - \beta K_{k-1}] - \delta_p q_k K_{k-1} + \Delta F_k. \tag{23}$$

If $(K_k - \beta K_{k-1})$ is the relevant measure for consumption of non-durables, the specification of the wealth constraint and the associated preference function imply that $(K_k - \beta K_{k-1})$ can be viewed just like any other good. (In this respect the term

 $-\delta_p q_k K_{k-1}$ can be interpreted as negative income.) This means that we can use the Hicks aggregation theorem for aggregation of these two goods into a Hicks composite good.

The application of the Hicks composite good theorem, cf. Deaton and Muell-bauer [9], requires that the price ratio q/p is constant over time. In the long run this assumption seems reasonable, and there are also arguments that justify this assumption as a reasonable approximation during the sample period, cf. section 7. In what follows we assume that the composite good theorem can be used.

3.3 The measurement of the composite good

For estimation purposes we need observations of Z. Let

$$Y_{k} \equiv C_{k} + \frac{q_{k}}{p_{k}} \left[K_{k} - (1 - \delta_{f}) K_{k-1} \right], \tag{24}$$

which is the cash flow from purchasing non-durables and durables measured in units of the price of non-durables. The composite good can now be expressed as

$$Z_k \equiv Y_k + \delta_p D_{k-1},\tag{25}$$

where

$$D_{k-1} \equiv \frac{q_k K_{k-1}}{p_k}$$

is the (real) market value of the stock of durables that was demanded at the end of the previous period, measured in the current prices. If $\delta_p = 0$, then $Z_k \equiv Y_k$, and specification (7) implies that households have preferences over the cash flow related to purchase of durables and non-durables.

In what follows we assume that we can observe D_{k-1} for at least one period. It remains to give an account of how we can observe Y_t . Since our data do not include reliable observations of this variable, we use the indirect approach. We have

$$Y_{k} = \frac{1}{p_{k}} \left[\sum_{j=f,m} w_{jk} H_{jk} + r_{k} F_{k-1} - I_{k} \left(w_{fk} H_{fk}, w_{mk} H_{mk}, r_{k} F_{k-1} \right) - \Delta F_{k} \right].$$
 (26)

Apart from the unknown habit persistence parameter δ_p , we observe all components of the composite good. We are left with the problem of obtaining reliable observations of the market value on durables, but we do not need to observe the consumption of non-durables, and the price and the physical stock of durables. It is also reasonable to assume that δ_p is small compared with Y, and measurement errors in D may then be less serious than in the standard framework presented in section 2. Specification (8)/(7), and the first order condition (13), then circumvent the problems related to obtaining reliable observations of the consumption of non-durables. Since durables are only one of the components of Z, we also believe that it reduces the problems related to the fixed costs in the demand for durables. In section 4 and 5 we show that δ_p can be treated as a parameter to be estimated.

3.4 Estimation strategy

Since the composite good, Z, with the exception of the unknown parameter, δ_p , can be treated as an ordinary good, the estimation approaches used for the standard life cycle model with intertemporal separability are relevant for the model outlined in the previous section. These approaches are surveyed in Kornstad [18], and we will now focus only on the use of the marginal rate of substitution functions, cf. MaCurdy [21] and Altonji [2].

According to economic theory, households adjust demand until the marginal rate of substitution between two arbitrary goods equals the relative price of the goods. For instance, if there are no binding constraints in the labour and the credit markets, we have

$$\frac{\partial U_{jt}/\partial L_{jt}}{\partial U_{zt}/\partial Z_t} = \frac{m_{jt}}{p_t}, \quad j = f, m.$$
 (27)

That is, in equilibrium the marginal rate of substitution between female/male leisure and the composite good is equal to the real after-tax marginal wage rate.

By choosing a particular class of lifetime preferences, the marginal utilities can be expressed in terms of the unknown parameters of the preferences and the (realized)

values of L_j and Z, and we obtain two specifications that can be used for estimation of the within-period utilities.

The possibility of bias in the estimates of labour supply functions from a subsample of workers that are unconstrained in the labour market is well known, cf. Heckman and MaCurdy [12]. According to Heckman [11], the bias can be viewed as an "omitted variables" bias.

Heckman assumes that it is possible to find a simple expression for the reduced form of the relationship to be estimated. This assumption simplifies the calculation of the omitted variable considerably, but the approach can not be easily modified to include structural equations. Except in very special cases it is impossible to find the reduced form equations of the lifetime paths for L_j and Z, so we ignore the possible selection problem.

Also notice that the marginal rate of substitution functions cannot be used to estimate the transformation function G. Using the Frisch demands for Z and the Euler equation (15) in the case of no binding constraints in the credit market, we find that

$$\frac{G'\partial U_{zt}/\partial Z_t}{p_t} = \frac{1}{1+\rho} \mathbf{E}_t \left[(1+R_{t+1}) \frac{G'\partial U_{zt+1}/\partial Z_{t+1}}{p_{t+1}} \right], \tag{28}$$

where G' is the partial derivative of G. Assuming e_{t+1} is the one-period forecast error, this expression implies

$$\frac{G'\partial U_{zt+1}/\partial Z_{t+1}}{p_{t+1}} = \frac{1+\rho}{1+R_{t+1}} \left[\frac{G'\partial U_{zt}/\partial Z_t}{p_t} \right] (1+e_{t+1}). \tag{29}$$

If expectations are rational, e_{t+1} is uncorrelated with

$$\frac{1+\rho}{1+R_{it+1}} \left[\frac{G'\partial U_{zt}/\partial Z_t}{p_t} \right], \tag{30}$$

and we have found an equation that can be used as a basis for estimation of the parameters of the G-function and the time preference rate ρ . In this situation, as with the estimation of equations based on the marginal rate of substitution functions, we face a possible selection problem if the relationship is estimated from a subsample of households that are unconstrained in the credit market.

Section 5.1 discusses the estimation procedure in more details.

4 A Particular Empirical Specification

In what follows we assume that the monotonic transformation function G is given by

$$G = \frac{U_{ik}^{\theta} - 1}{\theta},\tag{31}$$

and that (within-period) preferences are of the Box-Cox type

$$U_{ik} = \frac{(Z_{ik} + z_0)^{\sigma} - 1}{\sigma} + \Gamma_{ik} \frac{L_{fik}^{\gamma} - 1}{\gamma} + \Omega_{ik} \frac{L_{mik}^{\omega} - 1}{\omega}, \qquad (32)$$

where subscript i denotes household and θ , σ , γ and ω are unknown coefficients. The coefficients γ and ω determine the intertemporal substitution elasticities $1/(\gamma-1)$ and $1/(\omega-1)$, which measure the percentage change in the consumption of leisure in any two periods in response to a percentage change in the relative wage rate for those periods, cf. Heckman and MaCurdy [12]. The interpretation of σ is less straightforward, since $\beta \neq 0$, implying that Z is not an ordinary physical good.

If the within-period utilities are to be strictly concave, σ , γ and ω must all be less than one. Since the cash flow related to the purchases of durables and non-durables, as well as the parameter δ_p , are allowed to be negative, Z can be negative. If z_0 is zero and Z is negative, the (Box-Cox) utility of Z is undefined; in order to avoid this problem, we introduce the parameter z_0 . Notice also that since δ_p is unknown, the smallest value of $Y_t + \delta_p D_{t-1}$ can not be observed, and we (must) treat z_0 as a parameter to be estimated.

It is also assumed that the person- and age-specific modifiers of tastes, Γ_{ik} and Ω_{ik} , can be related to a vector of exogenous and observable consumer characteristics, X_{ik} and B_{ik} , and unmeasured characteristics, ε_{ik} and η_{ik} , according to $\Gamma_{ik} = \exp(X_{ik}\phi_x + \varepsilon_{ik})$ and $\Omega_{ik} = \exp(B_{ik}\phi_b + \eta_{ik})$. The error terms are needed since the econometrician cannot observe all components that influence preferences.

The first order conditions for leisure and consumption become

$$U_{it}^{\theta-1}(Z_{it}+z_0)^{\sigma-1}=\lambda_{it}p_t, (33)$$

$$U_{it}^{\theta-1}\Gamma_{it}L_{fit}^{\gamma-1} = \lambda_{it}m_{fit} + \alpha_{fit}$$
(34)

and

$$U_{it}^{\theta-1}\Omega_{it}L_{mit}^{\omega-1} = \lambda_{it}m_{mit} + \alpha_{mit}. \tag{35}$$

Assuming no binding constraints, either in the labour markets or in the credit market, the relationship between female leisure and the consumption good Z corresponding to equation (27), can be written

$$\ln L_{fit} = X_{it}a_1 + a_2 \ln \frac{m_{fit}}{p_t} + a_3 \ln (Y_{it} + z_0 + \delta_p D_{it-1}) + \nu_{it}, \tag{36}$$

where
$$a_1 = -\phi_x/(\gamma - 1)$$
, $a_2 = 1/(\gamma - 1)$, $a_3 = (\sigma - 1)/(\gamma - 1)$ and $\nu_{it} = -\varepsilon_{it}/(\gamma - 1)$.

The similar relationship between male leisure and Z can be written

$$\ln L_{mit} = B_{it}b_1 + b_2 \ln \frac{m_{mit}}{p_t} + b_3 \ln (Y_{it} + z_0 + \delta_p D_{it-1}) + \mu_{it}, \tag{37}$$

where
$$b_1 = -\phi_b/(\omega - 1), b_2 = 1/(\omega - 1), b_3 = (\sigma - 1)/(\omega - 1)$$
 and $\mu_{it} = -\eta_{it}/(\omega - 1)$.

It also follows that the Euler equation corresponding to equation (29) can be written

$$\ln\left[\left(1+R_{it+1}\right)\frac{p_t}{p_{t+1}}\left(\frac{Z_{it+1}+z_0}{Z_{it}+z_0}\right)^{\sigma-1}\right] = \ln(1+\rho) + (1-\theta)\ln\frac{U_{it+1}}{U_{it}} + \xi_{it+1}, \quad (38)$$
where $\xi_{it+1} = \ln(1+e_{it+1})$.

5 Estimating the subutilities

Ideally, equation (36), (37) and (38) should be estimated as a simultaneous equation system. We simplify, and use a two stage estimation procedure where we in the first stage estimate the subutilities, cf. equation (36) and (37), before we in the second stage estimate the monotonic transformation G, and the time preference rate ρ , cf. equation (38).

Comparing equation (36) and (37), we notice that the definitions of the parameters a_2 , a_3 , b_2 and b_3 yield four equations for the determination of the three coefficients γ , σ and ω . The model specification then implies a particular kind of parameter restriction across the two equations, which should be accounted for in

the estimation. Moreover, notice that these equations are non-linear in δ_p , and that there are endogenous right-hand-side variables. Estimation then requires a non-linear simultaneous equations procedure; and we apply a full-information maximum likelihood procedure.

First, however, we must choose the taste modifier variables for female and male leisure. We assume that preferences depend on age, and for the female, also on the size of the household. The taste modifier variable X then includes the number of children with age less than 21 years and the age of the female, while B is the age of the male. Even though the decision to have children depends on the behaviour of the household, we assume that these taste modifiers can be reasonably viewed as exogenous in the econometric specification, cf. also the discussion of the wage rate equations (39).

Estimation also requires a decision about what right-hand-side variables should be treated as endogenous from an econometric point of view, and a specification of the instrument equations for these variables. Notice that the logarithm of the after-tax wage rate can be decomposed into two components according to $\ln m_j = \ln(1 - \partial I/\partial(w_j H_j)) + \ln w_j$.

Since we do not allow for learning-by-doing, the household is price-taker in the labour market; this assumption does not necessarily mean that the econometrician should look at the (pre-tax) wage rates as exogenous, cf. Heckman [10]. If there are unobserved variables that influence the person's preferences for leisure, and that are correlated with the wage rate, the wage rate should be treated as endogenous in the empirical analysis. An example may be the comparison between a person who is very motivated for doing a good job at work, and one who is lazy and does not like to work. In this case we expect that the motivated person obtains a higher wage rate than the lazy one, even if we adjust for differences in education and work experience. It is also reasonable to believe that the motivated person has different preferences for work and leisure than the lazy one, and we have a situation where the unobserved variable influences both preferences and the wage rate. We may

label this kind of endogeneity "statistical endogeneity".

Many empirical analyses account for this endogeneity of the wage rate by introducing a separate wage equation. This equation typically includes the number of years of education and a measure of work experience as explanatory variables. Important explanatory variables such as motivation and ability are rarely included, and the specification suffers from the omission of these unobservables.

Another problem related to the specification of the wage rate equation is that estimation ideally requires individual life cycle data. If the wage rate equation is estimated from cross sectional data, the estimates also reflect differences in wage rates across cohorts, and the estimated wage equation will not be well-suited for predicting the wage rate for a particular person.

One may include person-specific fixed effects in order to control for unobserved variables. MaCurdy [20] assumes that wages follow a quadratic function in age with intercept and slope coefficients that depend on age-invariant characteristics of the consumer. This specification is then estimated using panel data for 10 years.

We do not have access to panel data that can be used for estimation of wage equations with person-specific fixed effects. Despite this fact, we find the argument for treating the pre-tax wage rate as endogenous quite convincing. Most empirical analyses also seem to adopt this assumption. Based on Heckman [10], and the findings in [8] by Dagsvik and Strøm, we assume that the logarithm of the wage rate is a linear function of work experience, experience squared and education, that is,

$$\ln w_{jit} = M_{jit}g_{aj} + v_{jit}, \quad j = f, m, \tag{39}$$

where M represents the consumer characteristics that are assumed to be exogenous from an econometric point of view, g_a is the corresponding vector of coefficients, and v is an error term. Education is measured in years, and work experience is measured as age minus the years of education minus the age at the beginning of education.

The discussion above and the fact that the wage rate equation is estimated from cross sectional data imply that one must take the greatest care in the interpretation

of the regression coefficients.

Since the marginal tax rate depends on the wage and interest incomes, and these depend on the behaviour of the household, the marginal tax rate is endogenous even for the household. To find the instrument for the logarithm of the after-tax marginal wage rate, it remains to determine the instrument for $\ln(1-\partial I/\partial(w_jH_j))$. We assume that the relevant explanatory variables are the same as for the wage equations, but in addition we include the number of children with age less than 21 years. This variable is included because both labour supply and the demand for durables and non-durables may be influenced by the size of the household. Wage incomes, financial savings and interest incomes and deductions are then influenced by the size of the household, and when the size of these variables is determined, the marginal tax rate is also determined. The instrument equations for the marginal tax rates are then

$$\ln\left(1 - \frac{\partial I}{\partial(w_j H_j)}\right) = B_{jit}g_{bj} + \epsilon_{jit}, \quad j = f, m, \tag{40}$$

where B is the vector of consumer characteristics, that is, experience, experience squared, education and the number of children with age less than 21 years, g_b is the corresponding vector of coefficients, and ϵ is the error term. Based on the above assumption, the vector B can be treated as exogenous in the estimation.

The cash flow related to purchases of durables and non-durables is endogenous for the household. We use education and education squared for the male as instrument variables for Y and add work experience for the female. Education is assumed to reflect the income potential of the household, but since the education of married females and males is (positively) correlated, we include only the education of the male in order to reduce the number of unknown coefficients. Using education squared, acknowledges that the consumption potential increases less than pre-tax wage incomes, since the tax system is progressive. Work experience is introduced as a substitute for age, since cash flow related to purchases of non-durables and, particularly durables, depends on age relative to the age when one finished education.

The instrument equation for the cash flow related to the purchases of durables and non-durables is then

$$Y_{it} = S_{it}g_s + \varsigma_{it}, \tag{41}$$

where S represents the consumer characteristics that are assumed to be exogenous from an econometric point of view, g_s is the corresponding vector of coefficients, and ς is an error term.

An alternative instrumental variable for Y would be gross or net income lagged one period, but since we find it reasonable to assume that there are omitted variables that are correlated with both lagged income and the error term in equation (36) and (37), we do not prefer this alternative.

The purchase price of non-durables is assumed equal to one in the estimation of the equations based on the marginal rate of substitution functions. It is also assumed that D_{t-1} is endogenous from an econometric point of view, although it is predetermined from the perspective of the household at age t. This decision is based on the assumption that there are omitted variables that are serially correlated and that influence preferences as well as D_{t-1} . The instruments used are female and male education, and an index of the number of inhabitants in the area where the household lives. Education is assumed to reflect income and consumption possibilities, while the number of inhabitants is introduced to try to account for the increase of the average price of housing as population density increases.

The instrument equation for the stock of durables is

$$D_{it-1} = J_{it}g_j + \psi_{it}, \tag{42}$$

where J is the vector of household characteristics that are assumed to be exogenous from an econometric point of view, g_j is the corresponding vector of coefficients, and ψ is an error term.

The complete model to be estimated consists of the structural equations (36) and (37) and the instrument equations (39) to (42). The properties of the error terms are as follows. Let $\mathbf{u}_{it} = (\nu_{it}, \mu_{it}, \nu_{fit}, \nu_{mit}, \epsilon_{fit}, \epsilon_{mit}, \varsigma_{it}, \psi_{it})'$ be the vector of

error terms for the simultaneous equation system. The subsequent analysis assumes that

$$\mathbf{E}\mathbf{u}_{it} = \mathbf{0}, \quad \forall i, t, \tag{43}$$

and that

$$E(\mathbf{u}_{it}\mathbf{u}_{it}') = \sum_{i}, \quad \forall i, t. \tag{44}$$

It is also assumed that \mathbf{u}_{it} is serially correlated, but uncorrelated between households. Assuming, in addition, that the components of \mathbf{u}_{it} follow a multivariate normal distribution, we can estimate the system by a full-information maximum likelihood procedure. All equations are identifiable, cf. Stewart and Wallis [29], and from the parameters of these equations it is possible to identify the parameters of preferences.

5.1 Estimating the monotonic transformation

Using the definition of U, the parameter estimates for σ , θ , ω , the taste modifier parameters ϕ_x and ϕ_b , and panel data observations of Y, q_tK_{t-1} , L_f , L_m as well as the taste modifier variables, it is possible to estimate $\ln(U_{t+1}/U_t)$ and $((Z_{t+1} + z_0)/(Z_t + z_0))^{\sigma-1}$ for use in the estimation of the Euler equation (38).

Estimating the Euler equation requires that we decide which households are unconstrained in the credit market, cf. (10). We do not have access to that kind of information, and for simplicity⁴ we assume that $\kappa = 0.9$. Since we cannot observe q_tK_t , but only q_tK_{t-1} , we also assume that households are unconstrained in the credit market if their net debt is less than 90 per cent of the market value of their durables measured in the prices of the next period, that is, if

$$-F_t < 0.9q_{t+1}K_t. (45)$$

The selection problem that may be involved by estimating the Euler equation from a subsample of unconstrained households, will be ignored.

⁴One reason may be that households cannot finance the purchase of durables without having any own capital. By choosing $\kappa = 0.9$, we approximate.

The properties of the error term $\xi_{it+1} = \ln(1 + e_{it+1})$ also require comment. Since ξ_{it+1} is a non-linear function of e_{it+1} , the assumption that $E_t e_{it+1} = 0$ does not imply that $E_t \xi_{it+1} = 0$. In what follows we assume that the (conditional) variance of e_{it+1} is constant across the sample, and that it is small. By approximating $\ln(1+e_{it+1})$ with a Taylor expansion, we then find that $E_t \xi_{it+1}$ is small, and that the (conditional) variance of ξ_{it+1} is constant. The Euler equation can then be estimated by a maximum likelihood procedure that treats the estimates for $\ln(U_{t+1}/U_t)$ as exogenous.

6 Data

The data are obtained from the panel data sections of the Income Distribution Survey 1988, 1989 and 1990, the Standard of Living Survey 1991, and an additional postal survey conducted as part of the Income Distribution Survey 1989. All data sets were collected and prepared by the Central Bureau of Statistics in Norway, and I have linked them on the basis of personal identification numbers.

The Income Distribution Surveys consist mainly of tax return data collected from taxation authorities. They yield quite detailed information about wage incomes, dividends, interest incomes and expenditures, financial wealth and debt, and income and property taxes. Transfer payments received, including child benefit, are also reported, and all these variables are used in the calculation of the cash flow related to the purchases of durables and non-durables. By linking two successive surveys, this survey also yields information about financial savings.

The Income Distribution Surveys also contain information about education, and the number and age of children. Given the tables of formal marginal tax rates, they also provide enough information for calculation of the marginal tax rates on wage and interest incomes. I have calculated the marginal tax rates in accordance with the set of regulations, and the possibility of separate taxation of the wage incomes of spouses is considered.

The ordinary Income Distribution Surveys also include the value of the stock of durables measured in the prices relevant for taxation, but these data underestimate stocks considerably. Instead I have linked the Income Distribution Survey 1989 with the additional postal survey, and the Income Distribution Survey 1990 with the Standard of Living Survey 1991. The postal survey and the Standard of Living Survey 1991 include questions about the market value of housing, cottages, cars and private motor boats in 1989 and 1990 respectively, and the variable $q_t K_{t-1}$ includes all these durables.

The postal survey and Standard of Living Survey 1991 also include information about working time in 1989 and 1990. The spouses are asked about their average working time per week and the number of weeks worked. Multiplying these two variables, yields an estimate for labour supply. Leisure is defined as 8736 hours (= total number of hours in a year) minus labour supply.

The (marginal) wage rate is measured as wage income divided by hours of labour supplied. If the worker gets overtime pay, and has much overtime, this wage rate will exceed his normal wage rate. In contrast, if the worker does not get overtime pay, and works much overtime, the observed wage rate underestimates the standard wage rate.

The data are combined in the following way. First we have linked the Income Distribution Survey 1989 and 1990 with the Standard of Living Survey 1991. (The Income Distribution Survey 1989 is included only to get data for financial savings through 1990.) Then we have linked the Income Distribution Survey 1988 and 1989 and the additional survey to the Income Distribution Survey 1989. (The Income Distribution Survey 1988 is included only to get data for financial savings through 1989.) From this last subsample we have excluded all persons who are included in the subsample from the first linkage. Finally, we combine these two subsamples into one sample.

The selection rules are as follows. Only married couples for which both spouses are between 30 and 55 years in 1990 are included. Those couples for which at

least one of the spouses has entrepreneurial income in excess of wage income are excluded. Couples for which one or both spouses receive disablement benefit and who have wage and entrepreneurial incomes less than 53 440 Norwegian kroner, are also excluded. Both spouses must work between 200 and 3500 hours⁵ a year, and the household must own their housing (not be tenants), to be included.

For some reason, the measurement method for the wage rate seems to bias the distribution of the wage rate, at least for females, cf. appendix 2. As a consequence, all couples for which the wage rate of the female exceeds 230 kroner or the wage rate of the male exceeds 270 kroner, are omitted from the analysis. The sample then consists of 327 observations. For further information about the sample, see summary statistics in table 1.

The pre-tax nominal interest rate is obtained by averaging over the quarterly observations of the private banks lending rents inclusive fees as they are presented in the statistics from the Norwegian Bank. This simplification may bias the estimation results since the (pre-tax nominal) interest rate do vary across households. The average consumer price index published by the Central Bureau of Statistics, is used to measure p_t/p_{t+1} . The average is based on monthly observations.

Estimating the Euler equation requires panel data for L_f , L_m , Y and D for two successive periods. The data are obtained by linking the Income Distribution Survey 1989 and 1990 with the Standard of Living Survey 1991, and the Income Distribution Survey 1988 and 1989 with the postal survey to the Income Distribution Survey 1989. These two subsamples are then linked, and the final sample consists of 229 households. In contrast to the procedure for the sample used in the estimation of the equations based on the marginal rate of substitution functions, only those households that are observed in both subsamples, are included.

⁵The method for the calculation of the wage rates implies that we observe the wage rate for all persons satisfying this labour supply constraint.

6.1 Some arguments in favour of Hicks aggregation

Complete household expenditure data for Norway are collected by the Central Bureau of Statistics. The data for non-durables are based on accounts of expenditures over two weeks, and expenditures for the whole year are found by multiplying these data by 26. The measurement errors introduced by this procedure can be huge, and we prefer to obtain the data from another data source.

This paper suggests aggregating C_k and $(K_k - \beta K_{k-1})$ using the Hicks composite good theorem. In light of price reductions for housing over the last years, the use of this theorem may be questioned. There are, however, some arguments that favour this assumption. The first is that the consumer price index, which we assume can be used to measure the price of non-durables⁶, includes expenditures related to housing as well. According to [7], the budget share of expenditures related to lighting and heating is about 5 per cent. The second point to notice is that durables include many variants, such as cars, private motorboats, furniture and kitchen and leisure equipment, and price changes for these goods have been less dramatic than for housing. The third point is that it seems reasonable to assume that the price path for durables is more or less similar to the price path for non-durables in the long run. So even if it can be questioned whether this approach is appropriate for the years included in this analysis, the approach may well be more defendable in other periods.

6.2 Measurement error

The possibility of measurement errors in some of the endogenous variables in the model makes the consequences of measurement errors of interest. The estimates of the equations based on the marginal rate of substitution functions are consistent only if the measurement error is of the classical errors in the variables type. This

⁶Unfortunately this index also includes prices of durables, and the observed price of non-durables may then be biased.

means that the observed endogenous variable A in the regression function is related to the true variable A_0 according to $A = A_0 + \varpi$, where ϖ is a randomly distributed error term that is distributed independently of all instrument variables. Given the results presented in appendix 2, the measurement problems related to durables, and the fact that the cash flow related to the purchases of durables and non-durables is observed indirectly through the wealth constraint, this is a very strong assumption.

The same issues of measurement errors arise in the discussion of the consequences of measurement errors in the estimation of the Euler equation (38). If there are measurement errors in equation (36) and (37), the error terms of these equations include measurement errors. This means that the estimate for U is biased, and when we use this estimate in the Euler equation, the measurement errors will be non-linear. In this case, we do not have a classical errors in the variables scheme, and the estimates will no longer be consistent.

7 Empirical Results

This section reports the estimates of the parameters of the subutility and monotonic transformation functions. Since we are not particularly interested in the results from the estimation of the instrument equations, these results are presented in appendix 1. That section also presents the estimates of the parameters of equation (36) and (37).

As it will become evident from this section, the estimation results are, with some exceptions, not very satisfying. One reason may be that we cannot model person-specific fixed effects in the wage equations as well as the other instrument equations, since we do not have access to panel data of sufficient length. Another reason may be the non-linearity in the two equations based on the marginal rate of substitution functions. We may also reasonably assume that treating Y_t and D_{t-1} as endogenous variables in the estimation enlarges the problems related to non-linearity. The fact that the samples include only 327 and 229 observations is also relevant. No matter

the reason, the estimation results should be viewed tentatively, and as part of ongoing research.

Since many of the estimates are quite imprecise in the sense that they are not statistically significant, it may be appropriate to clarify the implications of this result. The fact that a parameter is not statistically significant according to the standard T-test does not necessarily mean that we believe it is zero. If we have prior grounds for believing that it is different from zero (For instance, we may believe that females' preferences for leisure increase with an increase in the number of children.), we may continue to believe that even if the T-test does not support our original view. According to Wonnacott and Wonnacott [30], this conclusion is particularly relevant if the test is based on a small sample. The estimation results presented in this section, are based on a small sample.

Table 2 reports the estimates of the parameters of the subutilities. A priori we believe that the parameters of the taste modifiers are different from zero. The estimates of the intercepts of the taste modifiers are quite large, but none of them are statistically significant. We also notice that the estimate of the coefficient on the age variable is approximately the same for females as for males. The sign of these coefficients indicates that an increase in age decreases the preferences for leisure. This means that the marginal rate of substitution between female/male leisure and the composite good decreases, and female/male labour supply increases. These results should, however, be viewed in light of the fact that none of these estimates are determined precisely.

The sign of the estimate of the coefficient on children indicates that females' preferences for leisure increase with an increase in the number of children. The marginal rate of substitution between female leisure and the composite good then increases, and female labour supply decreases. The estimate is within about one standard error of the estimate of Heckman and MaCurdy [12], but it is imprecise.

The estimates for γ and ω are not significantly different from zero. We do not have knowledge of similar studies using Norwegian data, and cannot use this sort

of information for claiming that these parameters are different from zero. Both estimates are, however, consistent with the results in [31] by Aaberge, Dagsvik and Strøm, although they use a static and somewhat different model specification. The estimate for γ is also consistent with the results in Heckman and MaCurdy [13]. The implications of the finding that γ and ω are not significantly different from zero, are then ambiguous, in particular when we consider that the sample is small.

Both estimates are consistent with the assumption of a strict concave utility function. Since the absolute value of γ exceeds that of ω , the estimates also imply that male labour supply is more responsive to wage changes than female labour supply, cf. the first order conditions (34) and (35). This result seems to be in contrast with most empirical findings, which indicate that female labour supply is most responsive to wage changes.

If we look at the estimates of the parameters related to the preferences over the composite good, we notice that the standard errors of both z_0 and δ_p exceed the absolute value of the parameter estimates. This means that we cannot reject the null hypothesis that the psychological depreciation factor δ_p equals zero, and suggests that the relevant measure on the consumption of durables is $[K_t - (1 - \delta_f)K_{t-1}]$. The reason for introducing z_0 implies that z_0 should be greater than zero.

In contrast to all the other estimates, the estimate for σ is very precise. It is consistent with the assumption that preferences are strictly concave. Since there are no other studies that use our specification of the composite good Z, this result must be compared with the result of studies⁷ that use the standard consumption measure, cf. section 2.1. MaCurdy [21] includes imputed service flow for durables in the consumption measure, and finds that the parameter corresponding to σ is 0.34, with standard error equal to 0.21.

It remains to present the estimates from the estimation of the Euler equation (38), cf. table 3. Notice that the standard errors in this table do not account for

⁷The estimate for σ is very similar to the result in the (static) model of Aaberge, Dagsvik and Strøm [31].

the dependence of the variable $\ln(U_{t+1}/U_t)$ on estimated quantities.

In this case, the estimates for ρ and θ are both statistically significant. The estimate for θ implies that $\theta - 1$ is negative (= -0.112), and this result is consistent with risk averted households.

8 Summary

This paper formulates and estimates a life cycle model of labour supply and consumption of durables and non-durables for married couples in an environment of uncertainty and personal income taxes. While standard economic theory assumes that preferences are exogenously determined, we respond to the fact that many empirical studies have presented evidence in favour of utility being a relative concept, by specifying preferences subject to a reference standard. According to Boyer [6], the authors of these studies generally argue that although the assumption of given preferences can be reasonable in the short run, in a life cycle perspective it should be challenged. The lack of a well-established economic theory of habit formation, however, raises the question of how habit formation should be modelled. We introduce habit formation only in the demand for durables, and propose to use last year's demand for durables after deductions for physical and what we label psychological depreciation as the reference standard.

Since we do not have access to data for the consumption of non-durables, we aggregate non-durables and the good $(K_t - \beta K_{t-1})$ into a Hicks composite good. We then find a specification of the life cycle model of labour supply and consumption of durables and non-durables that can be estimated in the absence of data on the consumption of non-durables, and the price and the physical stock of durables. We argue that if the psychological depreciation parameter δ_p is small, this specification reduces the bias in the estimate of σ if there are measurement errors in observations of market value of the stock of durables.

In order to allow for more flexibility in lifetime utility, we decompose the period-

specific subutilities into one component that determines within-period allocations, and a monotonic transformation that influences the allocations between periods. The monotonic transformation also introduces a particular kind of non-separability in the within-period utilities.

In an empirical application of this specification, we assume that preferences are of the Box-Cox type, and suggest estimating the parameters of the subutilities by using the marginal rate of substitution functions between female/male leisure and the composite good. It is also possible to estimate the psychological depreciation parameter from these functions.

The monotonic transformation of the within-period utilities cannot be identified from these functions, but can be estimated by applying the Euler equation.

The data are obtained by linking the Income Distribution Surveys 1988, 1989 and 1990, the additional postal survey of the Income Distribution Survey 1989, and the Standard of Living Survey 1991. All data are linked on the basis of personal identification numbers. The estimates of θ and ω indicate that males' demand for leisure is more responsive to wage changes than females', but the estimation results should be viewed as tentative. None of these estimates are significantly different from zero. In contrast, the estimate of σ , which influences the preferences for consumption of durables and non-durables, is precise. All estimates are consistent with the assumption of strictly concave preferences.

The estimate of the psychological depreciation parameter is not statistically significant. This analysis then suggests that $[K_t - (1 - \delta_f)K_{t-1}]$ is the relevant measure of the consumption of durables in context of utility.

Finally, we estimate the Euler equation, but as we have argued, for this estimation to give consistent estimates, there is no room for measurement errors. It is then reasonable to assume that the estimates for θ and the time preference rate are biased.

Appendix 1

A Some further estimation results

The aim of this appendix is to give some further information about the estimates of the parameters of the subutilities by presenting the estimates for the parameters of the two marginal rate of substitution functions (36) and (37), and the instrument equations (39) to (42). (All the equations are estimated as a simultaneous equations system by FIML.)

Table 4 shows the estimates of the parameters of the marginal rate of substitution functions. With the exception of the intercepts, none of the variables have a t-statistic in excess of two.

We do not present the multiple correlation coefficients for the following reason. In the reduced form case, the multiple correlation coefficient, R^2 , measures the proportion of the original variance in the dependent variable that is explained by the regression as a whole. The marginal rate of substitution function, however, includes endogenous right-hand-side variables, and the meaning of R^2 becomes unclear. Further, the total sum of squares cannot be partitioned into explained and unexplained sums of squares for simultaneous equation estimates, since the residuals are not orthogonal to all of the explanatory variables. The residual sum of squares can then exceed the total sum of squares, and R^2 will be negative if it is calculated via the residual sum of squares.

Table 5 presents the results for the wage rate equations. All estimates are consistent with the results of Dagsvik and Strøm [8]. The parameters of the female wage rate equation are estimated less precisely than the parameters of the male equation.

Regarding the parameters of the instrument equation for $\ln(1 - \partial I/\partial(wH))$, cf. table 6, we notice that while the parameter related to the number of children with age less than 21 years is statistically significant for the female, it is determined rather imprecisely for the male. The estimates for the parameters related to experience,

experience squared and education for the females are within the estimates for the males plus or minus one standard error.

The lack of variables in our data that can reasonably be assumed exogenous from an econometric point of view, and the fact that the few we have access to vary little across the sample, make it hard to find instruments that can reproduce the large variation in cash flow related to the purchases of durables in the sample. From table 7 we also notice that the parameters related to male education and education squared, and to the age of the female relative to the age she completed her education, are all imprecisely determined.

Table 8 shows that the parameters of the instrument equation for the stock of durables are more precisely determined. The parameters related to female and male education are both statistically significant.

Appendix 2

A Measurement errors in the wage rate

The wage rate is measured by dividing wage incomes with labour supply, where labour supply is measured by multiplying average labour supply per week by the number of weeks worked, cf. section 7. For some reason this procedure seems to bias the distribution of the wage rate, at least for females, cf. table 9. This table shows the number of married females and males with wage rate in excess of 200 Norwegian kr and less than 800 kr, their average wage rate and average labour supply, for persons who are classified as employees, cf. section 7. The upper limit of the wage rate is chosen such that very extreme observations are omitted, and the data are obtained by linking the Income Distribution Survey 1990 with the Standard of Living Survey 1991. In order to increase the number of observations, it is now not conditioned that the families also are included in the Income Distribution Survey 1989, as is the case with the sample used in the empirical analysis, and the total sample consists of about 950 observations.

We notice that the average female wage rate is about 22 per cent higher than the average wage rate for males, and that the variance is much higher for females than for males. We also notice that average female labour supply is only about 400 hours, or about one fourth of average male labour supply. Even though the income effect of wage changes may reduce labour supply, these results seem unreasonable.

Taking also into consideration that the females in average are younger than the males and that they have shorter education, the wage rate for at least the females must be biased. As a consequence of these and some other results, all females with wage rate less than kr 40 and greater than kr 230 is omitted from the empirical analysis. For the males the corresponding limits are 50 and 270 kr.

Table 1: Summary statistics for the 327 married couples used for estimation of the equations based on the marginal rate of substitution functions.

	Average	Standard	Minimum	Maximum
		devation	value	value
Female leisure per year	7270	553	5616	8528
Male leisure per year	6717	353	5304	8466
Female gross wage rate, NOK				
per hour (among those who work)	94.7	30.2	43.1	219.8
Male gross wage rate, NOK				
per hour (among those who work)	117.7	33.8	67.3	266.7
Female marginal tax rate				
on wage incomes	0.407	0.090	0.231	0.62
Male marginal tax rate				
on wage incomes	0.523	0.095	0.266	0.62
Marginal tax rate on				
interest incomes/expenditures	0.394	0.064	0.260	0.456
Cash flow related to purchase				
of durables and non-durables	320361	227438	-41572	1498183
Stock of durables	1602300	879527	56000	5400000
Female education in years	10.9	2.2	8.0	17.5
Male education in years	11.7	2.8	8.0	19.0
Female age	41.5	6.3	30.0	55.0
Male age	43.9	6.2	30.0	55.0
Female experience in years	24.6	7.0	9.5	40.0
Male experience in years	26.2	7.1	8.5	41.0
Number of children 0-20 years	1.39	1.03	0.0	4.0
Index for inhabitants	3.42	1.30	1.0	5.0

Table 2: FIML estimates[†] for the subutility.

$\phi_{m{x}}$			ϕ_b			
Constant	Age	Children [‡]	γ	Constant	Age	ω
32.237	-0.0051	0.0565	-2.228	20.179	-0.0048	-0.8791
(26.496)	(0.0053)	(0.0408)	(2.999)	(11.950)	(0.0033)	(1.3955)

z_0	δ_p	σ
30241.9	0.0333	0.9558
(31029.4)	(0.0662)	(0.0359)

Table 3: ML estimation[†] of the monotonic transformation.

ρ	θ
0.0663	(0.8881)
(0.0026)	(0.0682)

[†] The asymptotic standard errors in parenthesis are not adjusted for the use of the instruments.

[†] Standard errors in parentheses.

† The number of children with age less than 21 years.

Table 4: FIML estimates[†] for the coefficients of the marginal rate of substitution functions.

	a_1					
Constant	Age	Children [‡]	a_2	a_3	z_0	δ_{p}
9.987	-0.0016	0.0175	-0.3098	0.0137	30241.9	0.0333
(1.083)	(0.0021)	(0.0103)	(0.2878)	(0.0142)	(31029.4)	(0.0662)

b :	_	
Constant Age		b_2
10.738	-0.0025	-0.5322
(1.627)	(0.0026)	(0.3952)

Table 5: FIML estimates[†] for the coefficients of the wage rate equation for females and males.

	Constant	Experience	Experience squared	Education
Female	3.920	0.0179	-0.00040	0.0373
	(0.237)	(0.0122)	(0.00023)	(0.0112)
Male	4.166	0.0188	-0.00037	0.0298
	(0.188)	(0.0105)	(0.00019)	(0.0067)

[†] Standard errors in parentheses.

[†] Standard errors in parentheses.

† The number of children with age less than 21 years.

Table 6: FIML estimates[†] for the coefficients of the instrument equation for $\ln(1 \partial I/\partial(wH)$) for females and males.

	Constant	Experience	Exp. squared	Education	Children [‡]
Female	0.0032	-0.0238	0.00044	-0.0245	0.0207
	(0.1286)	(0.0082)	(0.00017)	(0.0046)	(0.0106)
Male	-0.1222	-0.0216	0.00033	-0.0263	-0.00072
	(0.1634)	(0.0101)	(0.00019)	(0.0059)	(0.0039)

Table 7: FIML estimates[†] for the coefficients of the instrument equation for cash flow related to purchase of durables and non-durables.

Constant	Male education	Education squared	Female experience
-124399	63799.2	-1967.59	-693.798
(275512)	(40069.2)	(1479.99)	(2285.59)

[†] Standard errors in parentheses.

Table 8: FIML estimates for the coefficients of the instrument equation for the stock of durables.

Constant	Female	Male	Inhabitants
135258	66332.4	49135.4	48865.6
(306063)	(24659.0)	(22763.2)	(43467.2)

[†] Standard errors in parentheses.

[†] Standard errors in parentheses.

† The number of children with age less than 21 years.

Table 9: The distribution of female and male wage rate for employed persons with wage rate in excess of 200 NOK and less than 800 NOK.

	Female			Female Male			
Variable	Number	Mean	Variance	Number	Mean	Variance	
Wage rate	51	343.8	19197.0	88	281.7	10738.4	
Hours	51	403.7	60186.3	88	1642.3	676742.7	
Age	51	40.0	46.6	88	43.6	48.2	
Education	50	10.6	4.2	87	13.4	7.5	

[†] Labour supply measured in hours.

References

- [1] Alessie, R. and A. Kapteyn (1991), Habit Formation, Interdependent Preferences and Demographic Effects in the Almost Ideal Demand System, The Economic Journal, vol. 101, 404-419.
- [2] Altonji, J.G. (1986), Intertemporal Substitution in Labour Supply: Evidence from Micro Data, Journal of Political Economy, vol. 94, S176-S215.
- [3] Blundell, R. and I. Walker (1986), A Life-Cycle Consistent Empirical Model of Family Labour Supply Using Cross-Section Data, Review of Economic Studies, vol. LIII, 539-558.
- [4] Blundell, R. (1987), Econometric Approaches to the Specification of Life-Cycle Labour Supply and Commodity Demand Behaviour, Econometric Reviews, vol. 5, no. 1, 103-165.
- [5] Bover, O. (1991), Relaxing Intertemporal Separability: A Rational Habits Mode of Labor Supply Estimated from Panel Data, Journal of Labor Economics, vol. 9, no. 1, 85-100.
- [6] Boyer, O. (1983), Rational Demand and Expenditure Patterns under Habit Formation, Journal of Economic Theory, vol. 31, 27-53.
- [7] Central Bureau of Statistics, Oslo, Norway (1991), Konsumprisindeksen, Rapporter 91/8.
- [8] Dagsvik, J.K. and S. Strøm (1992), Labor Supply with Non-Convex Budget Sets, Hours Restrictions and Non-Pecuniary Job-Attributes, Discussion Paper no. 76, Central Bureau of Statistics, Oslo, Norway, 1-49.
- [9] Deaton, A. and J. Muellbauer (1989), Economics and Consumer Behaviour, Cambridge University Press.

- [10] Heckman, J.J. (1974), Shadow Prices, Market Wages, and Labour Supply, Econometrica, vol. 62, no. 4, 679-694.
- [11] Heckman, J.J. (1979), Sample Selection Bias as a Specification Error, Econometrica, vol. 47, 153-161.
- [12] Heckman, J.J. and T.E. MaCurdy (1980), A Life Cycle Model of Female Labour Supply, Review of Economic Studies, vol. 47, 47-74.
- [13] Heckman, J.J. and T.E. MaCurdy (1982), Corrigendum on A Life Cycle Model of Female Labour Supply, Review of Economic Studies, vol. XLIX, 659-660.
- [14] Hotz, V.J., F.E. Kydland and G.L. Sedlacek (1988), Intertemporal Preferences and Labor Supply, Econometrica, vol. 56, no. 2, 335-360.
- [15] Houthakker, H. and L. Taylor (1970), Consumer Demand in the United States, 2nd. ed., Harvard University Press, Cambridge, MA.
- [16] Johnson, T.R. and J.H. Pencavel (1984), Dynamic Hours of Work Functions for Husbands, Wives and Single Females, Econometrica, vol. 52, no. 2, 363-389.
- [17] Kapteyn, A. (1980), A Theory of Preference Formation, MRG Working Paper 7933, University of Southern California.
- [18] Kornstad, T. (1993), Empirical Approaches for Analysing Consumption and Labour Supply in a Life Cycle Perspective, Discussion Paper no. 95, Central Bureau of Statistics, Oslo, Norway, 1-56.
- [19] MaCurdy, T.E. (1980), An Intertemporal Analysis of Taxation and Work Disincentives, Working Papers in Economics no. E-80-4, The Hoover Institution, Stanford University, 1-53.
- [20] MaCurdy, T.E. (1981), An Empirical Model of Labor Supply in a Life-Cycle Setting, Journal of Political Economy, vol. 89, no. 6, 1059-1084.

- [21] MaCurdy, T.E. (1983), A Simple Scheme for Estimating an Intertemporal Model Labor Supply and Consumption in the Presence of Taxes and Uncertainty, International Economic Review, vol. 24, no. 2, 265-289.
- [22] MaCurdy, T.E. (1985), Interpreting Empirical Models of Labor Supply in an Intertemporal Framework with Uncertainty, Chapter 3, 111-155, J.J. Heckman and B. Singer, eds., Longitudinal Analysis of Labor Market Data, Cambridge University Press, London.
- [23] Mariger, R.P. (1987), A Life-Cycle Consumption Model with Liquidity Constraints: Theory and Empirical Results, Econometrica, vol. 55, no. 3, 533-557.
- [24] Muellbauer, J. (1986), Habits, Rationality and Myopia in the Life-Cycle Consumption Function, Discussion Paper no. 112, Centre for Economic Policy Research, London.
- [25] Phlips, L. (1972), A Dynamic Version of the Linear Expenditure Model, Review of Economics and Statistics, vol. 54, 450-458.
- [26] Phlips, L. (1974), Applied Consumption Analysis, Advanced Textbooks in Economics, vol. 5, Edited by C.J. Bliss and M.D. Intrilligator, North-Holland/American Elsevier.
- [27] Spinnewyn, F. (1981), Rational Habit Formation, European Economic Review, vol. 15, 91-109.
- [28] Stadt, H. van de, A. Kapteyn, S. van de Geer (1985), The Relativity of Utility: Evidence from Panel Data, The Review of Economics and Statistics, vol. LXVII, no. 2.
- [29] Stewart, M.B. and K.F. Wallis (1982), Introductory Econometrics, Basil Blackwell, Oxford.

- [30] Wonnacott, R.J. and T.H. Wonnacott (1985), Introductory Statistics, fourth edition, John Wiley and Sons, USA.
- [31] Aaberge, R., J.K. Dagsvik and S. Strøm (1990), Labor Supply, Income Distribution and Excess Burden of Personal Income Taxation in Norway, Discussion Paper no. 54, Central Bureau of Statistics, Oslo, Norway, 1-41.

ISSUED IN THE SERIES DISCUSSION PAPER

- No. 1 I. Aslaksen and O. Bjerkholt (1985): Certainty Equivalence Procedures in the Macroeconomic Planning of an Oil Economy.
- No. 3 E. Biøm (1985): On the Prediction of Population Totals from Sample surveys Based on Rotating Panels.
- No. 4 P. Frenger (1985): A Short Run Dynamic Equilibrium Model of the Norwegian Production Sectors.
- No. 5 I. Aslaksen and O. Bjerkholt (1985): Certainty Equivalence Procedures in Decision-Making under Uncertainty: An Empirical Application.
- No. 6 E. Biørn (1985): Depreciation Profiles and the User Cost of Capital.
- No. 7 P. Frenger (1985): A Directional Shadow Elasticity of Substitution.
- No. 8 S. Longva, L. Lorentsen and Ø. Olsen (1985): The Multi-Sectoral Model MSG-4, Formal Structure and Empirical Characteristics.
- No. 9 J. Fagerberg and G. Sollie (1985): The Method of Constant Market Shares Revisited.
- No. 10 E. Biørn (1985): Specification of Consumer Demand Models with Stochastic Elements in the Utility Function and the first Order Conditions.
- No. 11 E. Biørn, E. Holmøy and Ø. Olsen (1985): Gross and Net Capital, Productivity and the form of the Survival Function. Some Norwegian Evidence.
- No. 12 J.K. Dagsvik (1985): Markov Chains Generated by Maximizing Components of Multidimensional Extremal Processes.
- No. 13 E. Biørn, M. Jensen and M. Reymert (1985): KVARTS A Quarterly Model of the Norwegian Economy.

- No. 14 R. Aaberge (1986): On the Problem of Measuring Inequality.
- No. 15 A.-M. Jensen and T. Schweder (1986):
 The Engine of Fertility Influenced by
 Interbirth Employment.
- No. 16 E. Biørn (1986): Energy Price Changes, and Induced Scrapping and Revaluation of Capital A Putty-Clay Model.
- No. 17 E. Biørn and P. Frenger (1986): Expectations, Substitution, and Scrapping in a Putty-Clay Model.
- No. 18 R. Bergan, Å. Cappelen, S. Longva and N.M. Stølen (1986): MODAG A A Medium Term Annual Macroeconomic Model of the Norwegian Economy.
- No. 19 E. Biørn and H. Olsen (1986): A Generalized Single Equation Error Correction Model and its Application to Quarterly Data.
- No. 20 K.H. Alfsen, D.A. Hanson and S. Glomsrød (1986): Direct and Indirect Effects of reducing SO₂ Emissions: Experimental Calculations of the MSG-4E Model.
- No. 21 J.K. Dagsvik (1987): Econometric Analysis of Labor Supply in a Life Cycle Context with Uncertainty.
- No. 22 K.A. Brekke, E. Gjelsvik and B.H. Vatne (1987): A Dynamic Supply Side Game Applied to the European Gas Market.
- No. 23 S. Bartlett, J.K. Dagsvik, Ø. Olsen and S. Strøm (1987): Fuel Choice and the Demand for Natural Gas in Western European Households.
- No. 24 J.K. Dagsvik and R. Aaberge (1987): Stochastic Properties and Functional Forms of Life Cycle Models for Transitions into and out of Employment.
- No. 25 T.J. Klette (1987): Taxing or Subsidising an Exporting Industry.

- No. 26 K.J. Berger, O. Bjerkholt and Ø. Olsen (1987): What are the Options for non-OPEC Countries.
- No. 27 A. Aaheim (1987): Depletion of Large Gas Fields with Thin Oil Layers and Uncertain Stocks.
- No. 28 J.K. Dagsvik (1987): A Modification of Heckman's Two Stage Estimation Procedure that is Applicable when the Budget Set is Convex.
- No. 29 K. Berger, A. Cappelen and I. Svendsen (1988): Investment Booms in an Oil Economy -The Norwegian Case.
- No. 30 A. Rygh Swensen (1988): Estimating Change in a Proportion by Combining Measurements from a True and a Fallible Classifier.
- No. 31 J.K. Dagsvik (1988): The Continuous Generalized Extreme Value Model with Special Reference to Static Models of Labor Supply.
- No. 32 K. Berger, M. Hoel, S. Holden and Ø. Olsen (1988): The Oil Market as an Oligopoly.
- No. 33 I.A.K. Anderson, J.K. Dagsvik, S. Strøm and T. Wennemo (1988): Non-Convex Budget Set, Hours Restrictions and Labor Supply in Sweden.
- No. 34 E. Holmøy and Ø. Olsen (1988): A Note on Myopic Decision Rules in the Neoclassical Theory of Producer Behaviour, 1988.
- No. 35 E. Biørn and H. Olsen (1988): Production
 Demand Adjustment in Norwegian
 Manufacturing: A Quarterly Error Correction Model, 1988.
- No. 36 J.K. Dagsvik and S. Strøm (1988): A Labor Supply Model for Married Couples with Non-Convex Budget Sets and Latent Rationing, 1988.
- No. 37 T. Skoglund and A. Stokka (1988): Problems of Linking Single-Region and Multiregional Economic Models, 1988.

- No. 38 T.J. Klette (1988): The Norwegian Aluminium Industry, Electricity prices and Welfare, 1988.
- No. 39 I. Aslaksen, O. Bjerkholt and K.A. Brekke (1988): Optimal Sequencing of Hydroelectric and Thermal Power Generation under Energy Price Uncertainty and Demand Fluctuations, 1988.
- No. 40 O. Bjerkholt and K.A. Brekke (1988):
 Optimal Starting and Stopping Rules for
 Resource Depletion when Price is Exogenous and Stochastic, 1988.
- No. 41 J. Aasness, E. Biørn and T. Skjerpen (1988): Engel Functions, Panel Data and Latent Variables, 1988.
- No. 42 R. Aaberge, Ø. Kravdal and T. Wennemo (1989): Unobserved Heterogeneity in Models of Marriage Dissolution, 1989.
- No. 43 K.A. Mork, H.T. Mysen and Ø. Olsen (1989): Business Cycles and Oil Price Fluctuations: Some evidence for six OECD countries. 1989.
- No. 44 B. Bye, T. Bye and L. Lorentsen (1989): SIMEN. Studies of Industry, Environment and Energy towards 2000, 1989.
- No. 45 O. Bjerkholt, E. Gjelsvik and Ø. Olsen (1989): Gas Trade and Demand in Northwest Europe: Regulation, Bargaining and Competition.
- No. 46 L.S. Stambøl and K.Ø. Sørensen (1989): Migration Analysis and Regional Population Projections, 1989.
- No. 47 V. Christiansen (1990): A Note on the Short Run Versus Long Run Welfare Gain from a Tax Reform, 1990.
- No. 48 S. Glomsrød, H. Vennemo and T. Johnsen (1990): Stabilization of Emissions of CO₂: A Computable General Equilibrium Assessment, 1990.
- No. 49 J. Aasness (1990): Properties of Demand Functions for Linear Consumption Aggregates, 1990.

- No. 50 J.G. de Leon (1990): Empirical EDA Models to Fit and Project Time Series of Age-Specific Mortality Rates, 1990.
- No. 51 J.G. de Leon (1990): Recent Developments in Parity Progression Intensities in Norway. An Analysis Based on Population Register Data.
- No. 52 R. Aaberge and T. Wennemo (1990): Non-Stationary Inflow and Duration of Unemployment.
- No. 53 R. Aaberge, J.K. Dagsvik and S. Strøm (1990): Labor Supply, Income Distribution and Excess Burden of Personal Income Taxation in Sweden.
- No. 54 R. Aaberge, J.K. Dagsvik and S. Strøm (1990): Labor Supply, Income Distribution and Excess Burden of Personal Income Taxation in Norway.
- No. 55 H. Vennemo (1990): Optimal Taxation in Applied General Equilibrium Models Adopting the Armington Assumption.
- No. 56 N.M. Stølen (1990): Is there a NAIRU in Norway?
- No. 57 Å. Cappelen (1991): Macroeconomic Modelling: The Norwegian Experience.
- No. 58 J. Dagsvik and R. Aaberge (1991): Household Production, Consumption and Time Allocation in Peru.
- No. 59 R. Aaberge and J. Dagsvik (1991): Inequality in Distribution of Hours of Work and Consumption in Peru.
- No. 60 T.J. Klette (1991): On the Importance of R&D and Ownership for Productivity Growth. Evidence from Norwegian Micro-Data 1976-85.
- No. 61 K.H. Alfsen (1991): Use of Macroeconomic Models in Analysis of Environmental Problems in Norway and Consequences for Environmental Statistics.
- No. 62 H. Vennemo (1991): An Applied General Equilibrium Assessment of the Marginal Cost of Public Funds in Norway.

- No. 63 H. Vennemo (1991): The Marginal Cost of Public Funds: A Comment on the Literature.
- No. 64 A. Brendemoen and H. Vennemo (1991):
 A climate convention and the Norwegian economy: A CGE assessment.
- No. 65 K. A. Brekke (1991): Net National Product as a Welfare Indicator.
- No. 66 E. Bowitz and E. Storm (1991): Will Restrictive Demand Policy Improve Public Sector Balance?
- No. 67 Å. Cappelen (1991): MODAG. A Medium Term Macroeconomic Model of the Norwegian Economy.
- No. 68 B. Bye (1992): Modelling Consumers' Energy Demand.
- No. 69 K. H. Alfsen, A. Brendemoen and S. Glomsrød (1992): Benefits of Climate Policies: Some Tentative Calculations.
- No. 70 R. Aaberge, Xiaojie Chen, Jing Li and Xuezeng Li (1992): The Structure of Economic Inequality among Households Living in Urban Sichuan and Liaoning, 1990.
- No. 71 K.H. Alfsen, K.A. Brekke, F. Brunvoll, H. Lurås, K. Nyborg and H.W. Sæbø (1992): Environmental Indicators.
- No. 72 B. Bye and E. Holmøy (1992): Dynamic Equilibrium Adjustments to a Terms of Trade Disturbance
- No. 73 O. Aukrust (1992): The Scandinavian Contribution to National Accounting
- No. 74 J. Aasness, E, Eide and T. Skjerpen (1992): A Criminometric Study Using Panel Data and Latent Variables
- No. 75 R. Aaberge and Xuezeng Li (1992): The Trend in Income Inequality in Urban Sichuan and Liaoning, 1986-1990
- No. 76 J.K. Dagsvik and Steinar Strøm (1992): Labor Supply with Non-convex Budget Sets, Hours Restriction and Non-pecuniary Job-attributes

- No. 77 J.K. Dagsvik (1992): Intertemporal Discrete Choice, Random Tastes and Functional Form
- No. 78 H. Vennemo (1993): Tax Reforms when Utility is Composed of Additive Functions
- No. 79 J. K. Dagsvik (1993): Discrete and Continuous Choice, Max-stable Processes and Independence from Irrelevant Attributes.
- No. 80 J. K. Dagsvik (1993): How Large is the Class of Generalized Extreme Value Random Utility Models?
- No. 81 H. Birkelund, E. Gjelsvik, M. Aaserud (1993): Carbon/energy Taxes and the Energy Market in Western Europe
- No. 82 E. Bowitz (1993): Unemployment and the Growth in the Number of Recipients of Disability Benefits in Norway
- No. 83 L. Andreassen (1993): Theoretical and Econometric Modeling of Disequilibrium
- No. 84 K.A. Brekke (1993): Do Cost-Benefit Analyses favour Environmentalists?
- No. 85 L. Andreassen (1993): Demographic Forecasting with a Dynamic Stochastic Microsimulation Model
- No. 86 G.B. Asheim and K.A. Brekke (1993):
 Sustainability when Resource Management has Stochastic Consequences
- No. 87 O. Bjerkholt and Yu Zhu (1993): Living Conditions of Urban Chinese Households around 1990

- No. 88 R. Aaberge (1993): Theoretical Foundations of Lorenz Curve Orderings
- No. 89 J. Aasness, E. Biørn and T. Skjerpen (1993): Engel Functions, Panel Data, and Latent Variables with Detailed Results
- No. 90 Ingvild Svendsen (1993): Testing the REH. Using Norwegian Microeconomic Data (Will be issued later)
- No. 91 Einar Bowitz, Asbjørn Rødseth and Erik Storm (1993): Fiscal Expansion, the Budget Deficit and the Economy: Norway 1988-91 (Will be issued later)
- No. 92 Rolf Aaberge, Ugo Colombino and Steinar Strøm (1993): Labor Supply in Italy
- No. 93 Tor Jakob Klette (1993): Is Price Equal to Marginal Costs? An Integrated Study of Price-Cost Margins and Scale Economies among Norwegian Manufacturing Establishments 1975-90
- No. 94 John K. Dagsvik (1993): Choice Probabilities and Equilibrium Conditions in a matching Market with flexible Contracts
- No. 95 Tom Kornstad (1993): Empirical Approaches for Analysing Consumption and Labour Supply in a Life Cycle Perspective
- No. 96 Tom Kornstad (1993): An Empirical Life Cycle Model of Savings, Labour Supply and Consumption without Intertemporal Separability